Author’s response to reviews

Title: Model Consent Clauses for Rare Disease Research

Authors:

Minh Nguyen (thu.nguyen@mcgill.ca)
Jack Goldblatt (goldblattjack@gmail.com)
Rosario Isasi (RIsasi@miami.edu)
Marlene Jagut (marlene.jagut@irdirc.org)
Anneliene Jonker (aneliene.jonker@irdirc.org)
Petra Kaufmann (pk88@columbia.edu)
Laetitia Ouillade (louillade@afm-telethon.fr)
Fruszina Molnar-Gabor (fruzsina.molnar-gabor@adw.uni-heidelberg.de)
Mahsa Shabani (mahsa.shabani@kuleuven.be)
Eric Sid (eric.sid@nih.gov)
Anne Marie Tassé (anne-marie.tasse@mcgill.ca)
Durhane Wong-Rieger (durhane@sympatico.ca)
Bartha Knoppers (bartha.knoppers@mcgill.ca)

Version: 2 Date: 21 Jun 2019

Author’s response to reviews:

Editor Comments:

--Please include a Keywords section below your abstract listing three to ten keywords representing the main content of the article.

DONE

-- Unfortunately BMC does not accept boxes for publication. Please remove Box 1 and include the information in a different format.
"BOX" has been replaced by "TABLE"

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

DONE

-- Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

DONE

-- Please also add a section “Supplementary files” where you list the following information about your supplementary material:

- File name
- Title of data
- Description of data

DONE

Please ensure that all additional files are explicitly referred to in the main text.

-- Please submit the revised manuscript to the Ethics in Biomedical Research section.

DONE

-- At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

DONE